Author's response to reviews

Title: Health burden and economic impact of measles-related hospitalizations in Italy in 2002-2003.

Authors:

Antonietta Filia (antonietta.filia@iss.it)
Antonio Brenna (antonio.brenna@ciseditore.it)
Augusto Pana (pana@med.uniroma2.it)
Gianluca Maggio Cavallaro (maggiocavallaro@libero.it)
Marco Massari (marco.massari@iss.it)
Marta L Ciofi degli Atti (marta.ciofi@iss.it)

Version: 6 Date: 19 July 2007

Author's response to reviews: see over

Dear Editors,

Thank you for accepting our manuscript for publication in your journal. We have made the formatting changes requested and proofread the manuscript.

Kind regards,

Antonietta Filia MD
corresponding author (antonietta.filia@iss.it)